Author's response to reviews

Title: Empirical use of antibiotics and adjustment of empirical antibiotic therapies in a university hospital: a prospective observational study

Authors:

Julian Mettler (MettlerJ@uhbs.ch)
Mathew Simcock (SimcockM@uhbs.ch)
Pedram Sendi (Pedram.Sendi@unibas.ch)
Andreas F Widmer (WidmerA@uhbs.ch)
Roland Bingisser (BingisserR@uhbs.ch)
Manuel Battegay (BattegayM@uhbs.ch)
Ursula Fluckiger (FlueckigerU@uhbs.ch)
Stefano Bassetti (sbassetti_ol@spital.ktso.ch)

Version: 6 Date: 23 March 2007

Author's response to reviews: see over
Olten, March 23, 2007

MS: 8731830710305068 – Final version
„Empirical use of antibiotics and adjustment of empirical antibiotic therapies in a university hospital: a prospective observational study”

Dear Dr. Phillips,

Thank you for your e-mail confirming that BMC Infectious Diseases will publish our manuscript.

As you requested, we made the following changes to the format of our manuscript:

- The text “Research article” was removed from the title page.
- Section headings: all unnecessary capitals were removed.
- Competing interests: we wrote: “The authors declare that they have no competing interests”.

Sincerely,

Stefano Bassetti, M.D.

sbassetti_ol@spital.ktso.ch